January 31, 2023

Electronic Submission

Division of Dockets Management Food and Drug Administration Department of Health and Human Services 5630 Fishers Lane, Room 1061 Rockville, MD 20852

CITIZEN PETITION

This petition for administrative action is submitted on behalf of CAALM, the Coalition Advocating for Adequately Labeled Medicines ("Petitioner"), pursuant to 21 C.F.R. § 10.30 and related relevant provisions of the Federal Food, Drug, and Cosmetic Act or the Public Health Service Act, to request that the Commissioner of Food and Drugs (the "Commissioner") require that the sponsors of Comirnaty, Spikevax, Pfizer-BioNTech COVID-19 Vaccine, and Moderna COVID-19 Vaccine (collectively, "Pfizer and Moderna COVID-19 vaccines") amend current product labeling.

Incomplete, inaccurate, or misleading labeling of any medical product can negatively impact the health and safety of Americans, with global ramifications considering the international importance of FDA decisions. For these reasons, and due to the compelling need to ensure the safety and efficacy of any COVID-19 vaccine licensed by the FDA and to allow Petitioner the opportunity to seek emergency judicial relief should the instant Petition be denied, it is respectfully requested that FDA act on the instant Petition by April 30, 2023.

I. ACTIONS REQUESTED

Petitioner requests that the FDA amend current labelingⁱ of Pfizer and Moderna COVID-19 vaccines (for all authorized or approved indications and populations) in the following ways:

- Add language clarifying that phase III trials were not designed to determine and failed to provide substantial evidence of vaccine efficacy against SARS-CoV-2 transmission or death.
- 2. Add language clarifying that the immunobridging surrogate endpoint used in multiple authorized indications has not been validated to predict clinical efficacy.

ⁱ Labeling includes all Package Inserts (<u>Pfizer</u> & <u>Moderna</u>), all EUA Fact Sheets for Vaccination Providers (<u>Pfizer</u>, <u>Moderna</u>), and, as appropriate, all patient oriented materials such as EUA Fact Sheets for Recipients and Caregivers (<u>Pfizer</u>, <u>Moderna</u>) and Patient Package Inserts (<u>Moderna</u>).¹⁻⁹

- 3. Add safety and efficacy results data from manufacturer randomized trials of current bivalent boosters that reported results after EUA was granted.
- 4. Add a clear statement that FDA authorized a new Pfizer vaccine formulation containing Tris buffer without requiring clinical studies to evaluate efficacy, safety or bioequivalence to the formulation containing phosphate buffer.
- 5. Add a clear statement disclosing that a Pfizer phase III randomized trial in pregnant women (NCT04754594) was completed as of July 2022 but there have been no results reported.
- 6. Add a clear statement that Pfizer vaccine efficacy wanes after 2 months following dose 2 according to the Pfizer phase III randomized trial.
- 7. The following adverse event types should be added to the Adverse Reactions section of labeling:
 - a. multisystem inflammatory syndrome (MIS) in children;
 - b. pulmonary embolism;
 - c. sudden cardiac death;
 - d. neuropathic and autonomic disorders.
- 8. The following reproductive health and lactation related adverse event types should be added to the Adverse Reactions section of labeling:
 - a. decreased sperm concentration;
 - b. heavy menstrual bleeding;
 - c. detection of vaccine mRNA in breastmilk.
- 9. Add frequency data for clinical and subclinical myocarditis.
- 10. Labeling should present trial results on serious adverse events in tables with statistics, as is done for non-serious adverse events.

Petitioner also requests the FDA create a Medication Guide and communicate these labeling changes via a Dear Health Care Provider (DHCP) letter.

II. STATEMENT OF GROUNDS

Here, in the order as above, we set out the rationale for each requested action.

1. Add language clarifying that phase III trials were not designed to determine and failed to provide substantial evidence of vaccine efficacy against SARS-CoV-2 transmission or death.

In December 2020, when Emergency Use Authorizations (EUAs) were granted, FDA made clear that the clinical trials were not designed to evaluate--and data collected did not demonstrate--an effect against SARS-CoV-2 transmission.^{10,11}

Below is the wording from the FDA EUA review memo¹⁰ (the same statements are made for Moderna vaccine¹¹):

Vaccine effectiveness against asymptomatic infection

Data are limited to assess the effect of the vaccine against asymptomatic infection as measured by detection of the virus and/or detection of antibodies against non-vaccine antigens that would indicate infection rather than an immune response induced by the vaccine. Additional evaluations will be needed to assess the effect of the vaccine in preventing asymptomatic infection, including data from clinical trials and from the vaccine's use post-authorization.

Vaccine effectiveness against transmission of SARS-CoV-2

Data are limited to assess the effect of the vaccine against transmission of SARS-CoV-2 from individuals who are infected despite vaccination. Demonstrated high efficacy against symptomatic COVID-19 may translate to overall prevention of transmission in populations with high enough vaccine uptake, though it is possible that if efficacy against asymptomatic infection were lower than efficacy against symptomatic infection, asymptomatic cases in combination with

Vaccine effectiveness against mortality

A larger number of individuals at high risk of COVID-19 and higher attack rates would be needed to confirm efficacy of the vaccine against mortality. However, non-COVID vaccines (e.g., influenza) that are efficacious against disease have also been shown to prevent disease-associated death.¹²⁻¹⁵ Benefits in preventing death should be evaluated in large observational studies following authorization.

FDA EUA memo pp.50-51 (Dec 1, 2020)¹⁰

Months later, FDA's evaluation remained the same: efficacy against transmission remained unproven:

against hospitalization and death appears stable at this time. Remaining uncertainties regarding the clinical benefits of BNT162b2 in individuals 16 years and older include its level of protection against asymptomatic infection and transmission of SARS-CoV-2, including for the delta variant, durability of protection beyond 6-8 months (the current

FDA clinical review memo, p.98 (Aug 23, 2021)¹²

Today, FDA remains clear that efficacy against transmission is unproven. For <u>Pfizer</u> and <u>Moderna</u> vaccines,^{13,14} the FDA states:

A: Most vaccines that protect from viral illnesses also reduce transmission of the virus that causes the disease by those who are vaccinated. While it is hoped this will be the case, the scientific community does not yet know if the Pfizer-BioNTech COVID-19 Vaccine will reduce such transmission.

FDA "Pfizer-BioNTech COVID-19 Vaccine Frequently Asked Questions" (2023)13

While language in labeling that states what a product <u>has not</u> been proven to do is uncommon, it is necessary when caregivers and patients may inaccurately assume something that is untrue. For example, in 2000, <u>Roche promotional materials</u> stated that oseltamivir (Tamiflu) "reduces incidence of secondary complications (ie bacterial infections) by 45%" (an important pandemic

planning assumption).¹⁵ <u>FDA warned Roche</u> that this was a "misleading efficacy claim" that was "not supported by substantial evidence."¹⁶ Subsequently, <u>product labeling</u> (*Warnings and Precautions* section) was amended to state that "Tamiflu <u>has not</u> been shown" (emphasis added) to prevent serious complications of influenza.

5.3 Risk of Bacterial Infections

There is no evidence for efficacy of TAMIFLU in any illness caused by pathogens other than influenza viruses. Serious bacterial infections may begin with influenza-like symptoms or may coexist with or occur as complications during the course of influenza. TAMIFLU has not been shown to prevent such complications. Prescribers should be alert to the potential for secondary bacterial infections and treat them as appropriate.

Current FDA-approved TAMIFLU label (p.9)17

The situation is similar for Pfizer and Moderna vaccines. There is a widespread (but inaccurate) notion that efficacy against infection and transmission have been established by substantial evidence, and that these vaccines contribute to herd immunity. Inaccurate statements to this effect have been made by the <u>U.S. President</u>,¹⁸ <u>CDC Director</u>,¹⁹ and <u>former Director of the</u> <u>NIAID</u>,²⁰ and are directly implied in statements by <u>FDA</u>,²¹ <u>Pfizer</u>,²² and <u>Moderna</u>,²³ suggesting that widespread uptake of these vaccines would result in herd immunity.

President Biden 7/21/21: "you're not going to get COVID if you have these vaccinations."¹⁸

Dr. Anthony Fauci 5/16/21: "when you get vaccinated, you not only protect your own health ... you become a dead end to the virus."²⁰

<u>Dr. Rochelle Walensky 3/29/21</u>: "vaccinated people do not carry the virus, don't get sick, and that it's not just in the clinical trials but it's also in real world data."¹⁹

FDA 8/23/21: "Public health vaccination goals of immunizing 75% of the population (to achieve herd immunity) have not yet been achieved."¹²

<u>Pfizer 10/26/21</u>: "Maximizing the proportion of the population that is vaccinated is critically important to help reduce rates of infection, decrease transmission, prevent the emergence of new variants of concern, and hasten the end of the pandemic."²⁴

Moderna (2021-2023): "To safely achieve herd immunity against COVID-19, a large amount of a population needs to be vaccinated."²³

To remedy this situation, language clarifying that phase III trials were not designed to determine and failed to provide substantial evidence of vaccine efficacy against SARS-CoV-2 transmission or death must be added to labels.ⁱⁱ

ⁱⁱ Also see FDA labeling requirements (<u>21CFR201.57</u>) which states: "If there is a common belief that the drug may be effective for a certain use or if there is a common use of the drug for a condition, but the preponderance of evidence related to the use or condition shows that the drug is ineffective or that the therapeutic benefits of the product do not generally outweigh its risks, FDA may require that this section state that there is a lack of evidence that the drug is effective or safe for that use or condition."

2. Add language clarifying that the immunobridging surrogate endpoint used in multiple authorized indications has not been validated to predict clinical efficacy.

According to FDA's labeling requirements (21CFR201.57), drugs approved based on a surrogate endpoint must include "a succinct description of the limitations of usefulness of the drug and any uncertainty about anticipated clinical benefits."

FDA has granted multiple EUAs on the basis of trials that used an immunobridging primary efficacy endpoint (neutralizing antibody titers)--e.g., Pfizer EUAs for: 6 months to 4 year olds (see FDA memo p.6), 5 to 11 year olds (see FDA memo p.4), 12 to 15 year olds (see FDA memo p.5), booster #1 (see FDA memo p.4), booster #2 (see FDA memo p.2); and Moderna EUAs for 6 months to 17 year olds (see FDA memo p.10), booster #1 (see FDA memo p.6), and booster #2 (see FDA memo p.2).²⁵⁻³²

This immunobridging surrogate endpoint has not been validated to predict clinical efficacy.

Pfizer acknowledged th 2021, at the first Advise Committee meeting to discuss the first booste dose (dose 3), stating t it had not ascertained correlate of protection figure).

cknowledged this in	3	DR. KATHRIN JANSEN: German technology. I'm
t the first Advisory	4	just kidding. I just wanted to say that we actually
the first booster	5	looked in our breakthrough cases in our placebo-
ose 3), stating that	6	controlled phase III study and have compared the
ot ascertained a	7	antibody titers where we had the opportunity in
te of protection (see	8	individuals who got the disease versus the ones that
	9	didn't. And we were also unable to really come up with
	10	an antibody threshold. So I think it's probably a much
	11	more complex story and not just easily addressed with
FDA advisory committee	12	neutralizing antibodies. Thank you.
<u>meeting transcript</u> , p 243 (September 17	13	DR. JAMES HILDRETH: Thank you.
2021) ³³	14	DR. ARNOLD MONTO: That sounds reasonable.

FDA confirmed the lack of a validated surrogate endpoint in an April 2022 advisory committee meeting:

"There is not a clear, perfect, immune correlate of protection, and so we're using poor man's immune correlates of protection here -- or poor person's immune correlates of protection with antibody levels."34

FDA reiterated those concerns in a December 2022 JAMA commentary stressing the "Urgent Need for Next-Generation COVID-19 Vaccines":

"Therefore, unless correlates of protection that are strongly associated with duration of protection against COVID-19 can be identified, it is likely that rather than relying on immunobridging to infer vaccine effectiveness, large randomized clinical trials similar to the initial trials of the currently authorized or licensed vaccines for COVID-19 will be required to ascertain the effectiveness of these new vaccines."³⁵

While current labeling includes immunobridging efficacy results data, current labeling does not state that this endpoint has not been validated to predict clinical efficacy, as required by <u>21CFR201.57</u>. Labeling should be updated to reflect this.

3. Add safety and efficacy results data from manufacturer randomized trials of current bivalent boosters that reported results after EUA was granted.

On August 31, 2022, FDA granted Pfizer and Moderna EUAs for Bivalent (Original and Omicron BA.4/BA.5) vaccines.^{36,37} These authorizations were granted despite the fact that the new vaccine formulations had not completed any human testing.

While clinical data for the Pfizer-BioNTech COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5) are not yet available, FDA determined that for purposes of this EUA it is reasonable to assess the effectiveness and the known and potential benefits and risks of this bivalent vaccine based primarily on extrapolation of data from another bivalent vaccine, Bivalent BA.1, manufactured by the same process and containing original and Omicron BA.1 components, and extensive experience to date with the original Pfizer-BioNTech COVID-19 Vaccine. This

<u>FDA Decision Memo</u> p.5 (Aug 31, 2022)³⁶ Identical wording for Moderna in <u>FDA Decision Memo</u>, p.6 (Aug 31, 2022)³⁷

After the EUA was granted, in November 2022, <u>Pfizer</u> and <u>Moderna</u> both reported results in press releases of Phase 2/3 randomized trials of Bivalent (Original and Omicron BA.4/BA.5) boosters (<u>NCT05472038</u> and <u>NCT04927065</u>).^{38,39}

At present, labeling (*Section 18 Clinical Trial Results and Supporting Data for EUA*: <u>Pfizer p.36</u>, <u>Moderna p.36</u>) does not mention these trials or any other clinical trials of Bivalent (Original and Omicron BA.4/BA.5) vaccines;^{40,41} labeling should be updated to reflect current data.

4. Add a clear statement that FDA authorized a new Pfizer vaccine formulation containing Tris buffer without requiring clinical studies to evaluate efficacy, safety or bioequivalence to the formulation containing phosphate buffer.

Vaccine safety and efficacy data provided in labelingⁱⁱⁱ for children <u>6 months to 4 years</u> of age and <u>5 to 11 years</u> of age are based on Pfizer study C4591007 (<u>NCT04816643</u>),^{4,5} and data for

ⁱⁱⁱ Efficacy data from trials is presented in section 18 in Fact Sheets and section 14 in Package Inserts. Safety data from trials is provided in section 6 in Fact Sheets section 6.1 in Package Inserts.

ages <u>16 years and older</u> are based on Pfizer study C4591001 (<u>NCT04368728</u>) and BioNTech trial BNT162-01 (<u>NCT04380701</u>).¹

However, these trials all studied a formulation of the vaccine that differs from three of the four FDA authorized or approved formulations.

Specifically, these clinical trials all used a vaccine formulation containing a phosphate (PBS) buffer. However, the "gray cap," "maroon cap", and "orange cap" formulations in use contain a tromethamine (Tris) buffer.^{1,4,5}

The Pfizer vaccine containing a Tris buffer was first authorized in October 2021, for the EUA for 5 to 11 year olds. The <u>FDA decision memo</u> noted that the authorized vaccine was for a new formulation "containing tromethamine (Tris)/Sucrose buffer instead of the phosphate-buffered saline (PBS)/Sucrose buffer as used in the previous formulation."²⁶ No clinical trials were conducted to evaluate the efficacy, safety, or bioequivalence of the new product; only laboratory assessments were conducted.

Therefore, a clear statement should be added to labeling stating that the authorized or approved indications containing a Tris buffer is for a formulation that was not studied in these trials.

5. Add a clear statement disclosing that a Pfizer phase III randomized trial in pregnant women (NCT04754594) was completed as of July 2022 but there have been no results reported.

Pregnant women were excluded from the original clinical trials.^{42,43} In June 2021, a <u>New</u> <u>England Journal of Medicine editorial</u>⁴⁴ discussed the urgent need for trials in this population:

"The dearth of safety information about pregnancy, which existed at a time when thousands of pregnant women were grappling with decisions about vaccination, highlights the importance of recent efforts to enroll pregnant women in trials, including ongoing vaccine trials; a trial is currently under way to study the effects of the BNT162b2 vaccine in pregnant women and their infants (ClinicalTrials.gov number, NCT04754594)"⁴⁴

This Pfizer trial of pregnant women began in February 2021 (<u>NCT04754594</u>), but before results were available, <u>the CDC</u> and professional societies such as the American Congress of Obstetricians and Gynecologists (ACOG) <u>recommended</u> all pregnant women get vaccinated.^{45,46} The trial originally intended to enroll 4,000 women, but enrollment was inexplicably stopped in Q4 2021 with just 349 participants.

One year later, nothing has appeared in the literature: no publication, no preprint, no conference abstract.

Section 8.1 Pregnancy of current labeling discusses animal data, but omits mention that the existing trial in pregnant women has not yet reported results. This should be noted on the label. Once FDA receives the pregnancy trial data, results should be added to the label.

6. Add a clear statement that Pfizer vaccine efficacy wanes after 2 months following dose 2 according to the Pfizer phase III randomized trial.

Current labeling makes no mention of the data from Pfizer's phase 3 randomized trial showing (a) that efficacy is variable over time and (b) declines following an early peak.

An internal Pfizer report⁴⁷ shows that these results were available in April 2021, but they were not publicly disclosed until July 2021, in a <u>Pfizer preprint</u>.⁴⁸

In this same all-available (modified intention-to-treat) population, the estimated VE against all cases occurring \geq 7 days after Dose 2 was 91.2%. The estimated VE was 91.7% from \geq 11 days after Dose 1 to before Dose 2, 96.2% for cases occurring from \geq 7 days after Dose 2 to <2 months after Dose 2, 90.1% for the period from \geq 2 months to <4 months after Dose 2, and 83.7% for the period \geq 4 months after Dose 2.

Pfizer Interim 6-Month Report Body p.120 (April 29, 2021)47

Table 18.Vaccine Efficacy –Placebo-ControllePopulation	7 – First COVID-19 Occurrence After Dose 1 – Blinded led Follow-up Period – Dose 1 All-Available Efficacy						
The Dry	Vaccine Group (as Randomized)						
Ornar alth		BNT162b2 (30 µg) (N ^a =23040)		Placebo (N ^a =23037)			
Efficacy Endpoint Subgroup	n1 ^b	Surveillance Time ^c (n2 ^d)	n1 ^b	Surveillance Time ^c (n2 ^d)	VE (%)	(95% CI ^e)	
ts the the de							
First COVID-19 occurrence after Dose 1	131	8.412 (22505)	1034	8.124 (22434)	87.8	(85.3, 89.9)	
After Dose 1 to before Dose 2	46	1.339 (22505)	110	1.331 (22434)	58.4	(40.8, 71.2)	
After Dose 1 to <11 days after Dose 1	410	0.677 (22505)	50	0.675 (22434)	18.2	(-26.1, 47.3)	
≥11 Days after Dose 1 to before Dose 2	5	0.662 (22399)	60	0.656 (22369)	91.7	(79.6, 97.4)	
Dose 2 to 7 days after Dose 2	3	0.424 (22163)	35	0.422 (22057)	91.5	(72.9, 98.3)	
≥7 Days after Dose 2	82	6.649 (22132)	889	6.371 (22001)	91.2	(88.9, 93.0)	
≥7 days after Dose 2 to <2 Months after Dose 2	12	2.923 (22132)	312	2.884 (22001)	96.2	(93.3, 98.1)	
≥2 Months after Dose 2 to <4 Months after Dose 2	46	2.696 (20814)	449	2.593 (20344)	90.1	(86.6, 92.9)	
≥4 Months after Dose 2	24	1.030 (12670)	128	0.895 (11802)	83.7	(74.7, 89.9)	
Abbreviation: VE = vaccine efficacy.		On Jar		201			

Pfizer Interim 6-Month Report Body p.102 (April 29, 2021)47

7. The following adverse event types should be added to the Adverse Reactions section of labeling:

- multisystem inflammatory syndrome (MIS) in children;
- pulmonary embolism [Pfizer only];
- sudden cardiac death;
- neuropathic and autonomic disorders.

Since the introduction of COVID-19 vaccines, there has been a dramatic increase in reports to the Vaccine Adverse Event Report System (VAERS).

The volume of reports has coincided with an increase in published postmarketing safety studies.

A causal relationship <u>does not</u> need to be established before adding adverse events to the label that are detected in the postmarketing period. In fact, *Section 6.2 Postmarketing Experience* of the label <u>already states</u> that "Because these reactions are reported voluntarily from a population of uncertain size, it is not always possible to reliably estimate their frequency or establish a causal relationship to vaccine exposure."^{1,3}

The type, guality, and amount of evidence needed to justify adding a new adverse event type to labeling is not specified in FDA guidance. In the case of Janssen (J&J) COVID-19 vaccine, on April 13, 2021 the vaccine was "paused" pending the review of 6 cases of cerebral venous sinus thrombosis (CVST) with thrombocytopenia.⁴⁹ (All six reports came from VAERS; CDC's electronic health record real-time surveillance system Vaccine Safety Datalink did not detect a signal.⁵⁰) Ten days later, product labeling was amended to include a warning regarding this syndrome with a total of 15 confirmed cases of Thrombosis with Thrombocytopenia Syndrome (TTS) discovered within VAERS.⁵¹



<u>CDC Wonder search</u> on VAERS. Data current as of January 13, 2023 https://wonder.cdc.gov/controller/saved/D8/D321F831 **Multisystem inflammatory syndrome in children** should be added to *Section 6.2 Postmarketing Experience* of the label.

A <u>CDC and FDA authored study published in *Lancet Child & Adolescent Health* identified 6</u>

children, through August 31, 2021, who suffered multisystem inflammatory syndrome (MIS) following COVID-19 vaccination who

were without evidence of SARS-CoV-2 infection, had no alternative diagnosis, and no previous history of MIS-C.⁵² (See Figure.)

The CDC and FDA authors called for enhanced pharmacovigilance:

"Continued surveillance for MIS-C illness after COVID-19 vaccination is warranted, especially as paediatric COVID-19 vaccination is authorised for younger children, who comprise the highest proportion of MIS-C cases after SARS-CoV-2 infection. US providers are encouraged to report potential MIS-C cases after COVID-19 vaccination to VAERS."⁵²

	Total	Onset after first dose of COVID-19 vaccine	Onset after second dose of COVID-19 vaccine
MIS-C	21	11 (52%)	10 (48%)
With evidence of SARS-CoV-2 infection	15	10 (67%)	5 (33%)
Positive NAAT (past or recent)*	10	5 (50%)	5 (50%)
Negative NAAT and positive anti- nucleocapsid antibody test	5	4 (80%)	1 (20%)
Without evidence of SARS-CoV-2 infection†	6	1 (17%)	5 (83%)

Data are n or n (%). MIS-C=multisystem inflammatory syndrome in children. NAAT=nucleic acid amplification test. *Includes five individuals with positive SARS-CoV-2 NAAT before MIS-C illness, four with positive SARS-CoV-2 NAAT during MIS-C illness, and one with a positive SARS-CoV-2 NAAT before and during MIS-C illness. †Negative NAAT, negative anti-nucleocapsid antibody test, and positive anti-spike antibody test during MIS-C illness evaluation; these individuals did not have any reported positive NAAT results before MIS-C illness.

Table 2: SARS-CoV-2 laboratory testing in 21 individuals with MIS-C who had received a COVID-19 vaccine

* * *

Pulmonary embolism should be added to *Section 6.2 Postmarketing Experience* of Pfizer vaccine labeling.

An <u>FDA-authored postmarketing study published in *Vaccine* detected a signal for this serious adverse event.⁵³</u>

Findings: Four outcomes met the threshold for a statistical signal following BNT162b2 vaccination including pulmonary embolism (PE; RR = 1.54), acute myocardial infarction (AMI; RR = 1.42), disseminated intravascular coagulation (DIC; RR = 1.91), and immune thrombocytopenia (ITP; RR = 1.44). After further evaluation, only the RR for PE still met the statistical threshold for a signal; however, the RRs for AMI, DIC, and ITP no longer did. No statistical signals were identified following vaccination with either the mRNA-1273 or Ad26 COV2.S vaccines.

FDA study (Wong et al. 2022)⁵³

Although these findings were made between February and April 2021, they have not yet been added to product labeling.

* * *

Sudden cardiac death should be added to Section 6.2 Postmarketing Experience of the label.

Multiple autopsy-based published studies have reported on lethal vaccination-associated myocarditis.^{54–57} In one paper, the <u>Chief Medical Examiner of Connecticut and colleagues</u> published on 2 adolescents who died following the second dose of Pfizer vaccine, finding "features resembling a catecholamine-induced injury, not typical myocarditis pathology."⁵⁶ In

another autopsy study published in *Clinical Research in Cardiology*, researchers reported on all autopsies (n=35) performed at the Institute of Pathology, Heidelberg University Hospital in which death occurred within 20 days of COVID-19 vaccination. mRNA vaccines (Pfizer and Moderna) were judged as the "likely or possible" cause of death for 5 individuals.⁵⁷

Here, we describe the cardiac autopsy findings in five persons who have died unexpectedly within seven days following anti-SARS-CoV-2-vaccination, with vaccine-induced myocardial inflammation representing the likely or possible cause of death. Our findings establish the histological phenotype of lethal vaccination-associated myocarditis.

Schwab et al. Clin Res Cardiol. 2022 Nov 27:1–10.

* * *

Neuropathic and autonomic disorders should be added to *Section 6.2 Postmarketing Experience* of the label.

In May 2022, <u>an NIH study</u> reported on 23 patients reporting new neuropathic symptoms following vaccination.⁵⁸

Conclusions: This observational study suggests that a variety of neuropathic symptoms may manifest after SARS-CoV-2 vaccinations and in some patients might be an immune-mediated process.

Safavi et al. medRxiv [Preprint] (May 17, 2022)58

In December 2022, a <u>study published in *Nature Cardiovascular Research*</u> reported elevated risk of postural orthostatic tachycardia syndrome (POTS) and dysautonomia.⁵⁹



Figure 1 in Kwan et al. Nat Cardiovasc Res 2022; 1:1187–1194.

A <u>linked editorial</u>⁶⁰ discussing the study, stated that the research:

"undeniably establishes POTS and dysautonomia in general as adverse events after vaccination that should be recognized and investigated as other well-accepted postvaccination syndromes, such as Guillain–Barre syndrome and acute disseminated encephalomyelitis."⁶⁰

8. The following reproductive health and lactation related adverse event types should be added to the Adverse Reactions section of labeling:

- a. decreased sperm concentration [Pfizer only];
- b. heavy menstrual bleeding;
- c. detection of vaccine mRNA in breastmilk.

Decreased sperm concentration should be added to *Section 6.2 Postmarketing Experience* of the Pfizer label.

Section 13.1 Carcinogenesis, Mutagenesis, Impairment of Fertility of the Pfizer and Moderna labels states that neither vaccine has been studied for the potential to impair male fertility.^{1,3}

13.1 Carcinogenesis, Mutagenesis, Impairment of Fertility

COMIRNATY has not been evaluated for the potential to cause carcinogenicity, genotoxicity, or impairment of male fertility. In a developmental toxicity study in rats with COMIRNATY there were no vaccine-related effects on female fertility *[see Use in Specific Populations (8.1)]*.

FDA-approved <u>Pfizer vaccine label</u>, p.21 (Similar language in <u>Moderna vaccine label</u> p.13)

A postmarketing study of sperm donors who had received Pfizer vaccine in Israel, published in <u>Andrology</u>, however, found that vaccination temporarily impairs semen concentration and total motile count among

semen donors.⁶¹

The temporary decline in sperm concentration and total motile count should be added to product labeling.

In conclusion, in this longitudinal multicenter study, we found a selective temporary decline of sperm concentration and total motile count 3 months post-vaccination followed by recovery among SD. While on first look, these results may seem concerning, from a clinical perspective they confirm previous reports regarding

Gat et al. Andrology. 2022 Sep;10(6):1016-1022.

* * *

Heavy menstrual bleeding should be added to *Section 6.2 Postmarketing Experience* of the label.

In October 2022, the European Medicines Agency (EMA) has advised that heavy menstrual be added to product labeling.

Comirnaty and Spikevax: heavy menstrual bleeding added as a side effect

The <u>PRAC</u> has recommended that heavy menstrual bleeding should be added to the <u>product</u> <u>information</u> as a side effect of unknown frequency of the mRNA COVID-19 vaccines Comirnaty and Spikevax.

European Medicines Agency website (Oct 28, 2022)62

An <u>NIH-funded study published in *BMJ Medicine*</u>, and reporting by the <u>New York Times</u> and <u>Washington Post</u>, have documented additional menstrual irregularities.^{63–65}

At present, Pfizer and Moderna product labeling does not mention anything related to menstrual irregularities.

* * *

Detection of vaccine mRNA in breastmilk should be added to Section 8.2 Lactation of the label.

The Pfizer label states: "It is not known whether COMIRNATY is excreted in human milk."¹ (Identical language appears for Moderna's SPIKEVAX.³)

However, a study published in <u>JAMA Pediatrics</u> has documented the presence of vaccine mRNA in breast milk for at least 48 hours after maternal vaccination.⁶⁶

Table 2. Detection of Vaccine RNA in Whole Expressed Breast Milk and Extracellular Vesicles in 5 Patients at Various Time Points Postvaccination					
Participant No.	Vaccine type	Time points of vaccine mRNA detection in EBM	Concentration of vaccine mRNA detected in whole milk ^a	Concentration of vaccine mRNA detected in EBM EVs ^a	Abbreviation: EBM, expressed breast
4	BNT162b2	27-h ^b Sample	Not detected	14.01 pg/mL	mRNA, messenger RNA.
6	mRNA-1273	27-h and 42-h ^b Samples	11.7 pg/mL	16.78 pg/mL	^a Units for concentration are
7	BNT162b2	37-h ^b Sample	Not detected	4.69 pg/mL	picogram of mRNA per milliliter
8	BNT162b2	1-h and 3-h ^b Samples	1.3 pg/mL	6.77 pg/mL	of whole milk equivalent.
10	mRNA-1273	45-h ^b Sample	2.5 pg/mL	2.13 pg/mL	Sample used for vaccine mRNA concentration detection.

Table 2 in Hanna et al. JAMA Pediatr. 2022;176(12):1268–1270.

9. Add frequency data for clinical and subclinical myocarditis.

Current labeling provides no information on the frequency of myocarditis or pericarditis. Labels should contain a range of rates that have been reported in the literature, and should stratify estimates by risk factors (notably, age and sex).⁶⁷ For example, a prospective cohort study of 301 adolescents in Thailand measured cardiovascular outcomes following dose 2 of Pfizer vaccine. Myopericarditis was confirmed in one patient (1/301), two patients had suspected pericarditis, and 4 patients had suspected subclinical myocarditis.⁶⁸ A Hong-Kong cohort study of males 12-17 years old reported increased risk of myocarditis/pericarditis at 1/2700 following dose 2 of Pfizer vaccine.⁶⁹ And an FDA analysis estimated myocarditis/pericarditis increased risk at 1/5000 among vaccinated males 16-17 years old.²⁸

In addition, the label states: "Information is not yet available about potential long-term sequelae."^{1,3}

This statement is outdated, as studies now exist. For example, a <u>CDC study in The</u> <u>Lancet Child and Adolescent</u> <u>Health</u> found 50% (178/357) of adolescents with postvaccine myocarditis had \geq 1 symptom lasting at least 90 days from onset of myocarditis.⁷⁰

In the 2 weeks before the survey date, 178 (50%) of 357 patients reported having at least one symptom that might occur with myocarditis (chest pain or discomfort, fatigue, shortness of breath, or palpitations). Patients who were not considered recovered from myocarditis more frequently reported fatigue than did patients who were considered recovered (12 [43%] *vs* 40 [21%]; p=0.018; table 1). By contrast, based on the last patient encounter, health-care providers reported that 62 (16%) of 393 patients at least one symptom that might occur with myocarditis (table 1).

Kracalik et al. Lancet Child Adolesc Health. 2022 Nov;6(11):788-798.

10. Labeling should present trial results on serious adverse events in tables with statistics, as is done for non-serious adverse events.

The potential benefits of vaccination vary by population and over time. To enable providers to ensure benefits outweigh risks for individual patients, labeling should clearly convey the risk of serious adverse events.

Currently, product labeling and the EUA Fact Sheet report commonly experienced adverse events (redness, swelling, pain at injection site) in tabular format, as is expected.

However serious adverse events are <u>not</u> presented in tabular format. This prevents easy understanding of risk. FDA should require SAE be presented in tabular format e.g. similar to how <u>Fraiman et al</u>. presented the RCT data.⁷¹

	Total events (events per	r 10,000 participants) ^a	Risk difference	Risk ratio (95 % CI) ^e
Trial	Vaccine	Placebo	per 10,000 participants (95 % CI) ^e	
Serious advers	e events			
Pfizer ^b	127 (67.5)	93 (49.5)	18.0 (1.2 to 34.9)	1.36 (1.02 to 1.83)
Modernac, d	206 (135.7)	195 (128.6)	7.1 (-23.2 to 37.4)	1.06 (0.84 to 1.33)
Combined ^f	333 (98.0)	288 (84.8)	13.2 (-3.2 to 29.6)	1.16 (0.97 to 1.39)

Table 2 in Fraiman et al. Serious adverse events of special interest following mRNA COVID-19 vaccinationin randomized trials in adults. Vaccine. 2022 Sep 22;40(40):5798-5805.

III. ENVIRONMENT IMPACT

The petitioner hereby states that the relief requested in this petition will have no environmental impact and therefore an environmental assessment is not required under 21 C.F.R. Sections 25.30 and 25.31.

IV. ECONOMIC IMPACT

Economic impact information will be submitted upon request of the commissioner.

V. CERTIFICATION

The undersigned certifies that, to the best knowledge and belief of the undersigned, this petition includes all information and views on which the petition relies, and that it includes representative data and information known to the petitioner which are unfavorable to the petition.

Respectfully submitted,

Linda Wastila

Linda Wastila, BSPharm, MSPH, PhD Representative Coalition Advocating for Adequately Labeled Medicines (CAALM)

Coalition Advocating for Adequately Labeled Medicines (CAALM), current members as of January 31, 2023:

G. Caleb Alexander, MD, MS⁺

Professor, Department of Epidemiology Johns Hopkins Bloomberg School of Public Health Baltimore, Maryland [†] Dr. Alexander's organizational affiliation is included for identification purposes only.

Chris D'Adamo, PhD⁺

Director, Center for Integrative Medicine Assistant Professor, Department of Family and Community Medicine & Department of Epidemiology and Public Health University of Maryland School of Medicine Baltimore, Maryland

[†] Dr. D'Adamo's organizational affiliation is included for identification purposes only.

Peter Doshi, PhD[†]

Associate Professor, Department of Practice, Sciences, and Health Outcomes Research University of Maryland School of Pharmacy Baltimore, Maryland [†] Dr. Doshi's organizational affiliation is included for identification purposes only.

Tianjing Li, MD, MHS, PhD⁺

Associate Professor Department of Ophthalmology Department of Epidemiology University of Colorado Anschutz Medical Campus Aurora, Colorado [†] Dr. Li's organizational affiliation is included for identification purposes only.

Martin A. Makary M.D., M.P.H.[†]

Professor, Johns Hopkins School of Medicine Chief, Islet Transplant Surgery Johns Hopkins Hospital Baltimore, Maryland [†] Dr. Makary's organizational affiliation is included for identification purposes only.

Vinay Prasad, MD MPH[†]

Professor, Department of Epidemiology and Biostatistics University of California San Francisco San Francisco, California [†] Dr. Prasad's organizational affiliation is included for identification purposes only.

Linda Wastila, BSPharm, MSPH, PhD^{*†}

Professor, Department of Practice, Sciences, and Health Outcomes Research University of Maryland School of Pharmacy 220 Arch Street, Baltimore, Maryland 21201 * Dr. Wastila is serving as the Representative of CAALM * Dr. Wastila's organizational affiliation is included for identification purposes only.

Patrick Whelan, MD PhD⁺

Associate Clinical Professor of Pediatrics David Geffen School of Medicine at UCLA Los Angeles, California [†] Dr. Whelan's organizational affiliation is included for identification purposes only.

Kim Witczak

President/Co-Founder Woodymatters Minneapolis, Minnesota

References

- 1. Pfizer Inc. Full Prescribing Information, COMIRNATY Gray Cap for individuals 12 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/154834/download</u>
- 2. Pfizer Inc. Full Prescribing Information, COMIRNATY Purple Cap for individuals 12 years of age and older [Internet]. 2022. Available from: https://www.fda.gov/media/151707/download
- ModernaTx, Inc. Full Prescribing Information, SPIKEVAX for individuals 18 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/155675/download</u>
- Pfizer Inc. EUA Fact Sheet for Healthcare Providers, Pfizer-BioNTech COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5), Maroon Cap for individuals 6 months through 4 years of age [Internet]. 2022. Available from: <u>https://www.fda.gov/media/159312/download</u>
- Pfizer Inc. EUA Fact Sheet for Healthcare Providers, Pfizer-BioNTech COVID-19 Vaccine, Orange Cap for individuals age 5 years to <12 years [Internet]. 2022. Available from: <u>https://www.fda.gov/media/153714/download</u>
- Pfizer Inc. EUA Fact Sheet for Recipients and Caregivers, Pfizer-BioNTech COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5), for individuals 12 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/153716/download</u>
- Pfizer Inc. EUA Fact Sheet for Healthcare Providers, Pfizer-BioNTech COVID-19 Vaccine, Gray Cap for individuals 12 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/153715/download</u>
- ModernaTx Inc. EUA Fact Sheet for Healthcare Providers, Moderna COVID-19 Vaccine, red cap for individuals 12 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/157233/download</u>
- ModernaTx Inc. EUA Fact Sheet for Recipients and Caregivers, SPIKEVAX and Moderna COVID-19 Vaccine, for individuals 6 years of age and older [Internet]. 2022. Available from: <u>https://www.fda.gov/media/159310/download</u>
- Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19 Vaccine for individuals 16 years of age and older [Internet]. 2020 Dec [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/144416/download</u>
- 11. Food and Drug Administration. FDA EUA review memorandum for Moderna COVID-19 Vaccine [Internet]. 2020 Dec [cited 2022 Feb 21]. Available from: <u>https://www.fda.gov/media/144673/download</u>
- 12. Wollersheim S, Schwartz A. FDA BLA Clinical Review Memorandum, COMIRNATY [Internet]. 2021 Aug. Available from: <u>https://www.fda.gov/media/152256/download</u>
- Food and Drug Administration. Pfizer-BioNTech COVID-19 Vaccine Frequently Asked Questions [Internet]. U.S. Food and Drug Administration. FDA; 2022 [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/emergency-preparedness-and-response/coronavirus-disease-2019-covid-19/pfizer-biontech-covid-19-vaccine-frequently-asked-questions</u>

- 14. Food and Drug Administration. Q&A for Spikevax (COVID-19 Vaccine mRNA) [Internet]. U.S. Food and Drug Administration. FDA; 2022 [cited 2023 Jan 22]. Available from: https://www.fda.gov/vaccines-blood-biologics/qa-spikevax-covid-19-vaccine-mrna
- 15. Roche. Tame-the-flu with Tamiflu [Internet]. 2000 [cited 2023 Jan 22]. Available from: <u>http://web.archive.org/web/20160125202602/http://www.fda.gov/downloads/Drugs/GuidanceCo</u> <u>mplianceRegulatoryInformation/EnforcementActivitiesbyFDA/WarningLettersandNoticeofViolation</u> <u>LetterstoPharmaceuticalCompanies/UCM165845.pdf</u>
- 16. Food and Drug Administration. 2000. Available from: <u>http://web.archive.org/web/20160124090229/http://www.fda.gov/downloads/Drugs/GuidanceCo</u> <u>mplianceRegulatoryInformation/EnforcementActivitiesbyFDA/WarningLettersandNoticeofViolation</u> <u>LetterstoPharmaceuticalCompanies/UCM166329.pdf</u>
- 17. Genentech. Full Prescribing Information, TAMIFLU [Internet]. 2019. Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2019/021087s071,021246s054lbl.pdf
- Biden J. Remarks by president Biden in a CNN Town Hall with don lemon [Internet]. The White House. 2021 [cited 2023 Jan 22]. Available from: <u>https://www.whitehouse.gov/briefing-</u> <u>room/speeches-remarks/2021/07/22/remarks-by-president-biden-in-a-cnn-town-hall-with-donlemon/</u>
- Universal NBC. Transcript: The Rachel maddow show, 3/29/21 [Internet]. MSNBC. 2021 [cited 2023 Jan 22]. Available from: <u>https://www.msnbc.com/transcripts/transcript-rachel-maddow-show-3-29-21-n1262442</u>
- 20. Transcript: Dr. Anthony Fauci on "Face the Nation," May 16, 2021 [Internet]. CBS News. 2021 [cited 2023 Jan 22]. Available from: <u>https://www.cbsnews.com/news/transcript-dr-anthony-fauci-face-the-nation-05-16-2021/</u>
- 21. Zhang R, Goswami J. FDA BLA Clinical Review Memorandum, SPIKEVAX [Internet]. 2022 Jan. Available from: <u>https://www.fda.gov/media/156342/download</u>
- 22. Pfizer Inc. The ability to vaccinate at speed to gain herd immunity and stop transmission is our highest priority. There is a lot of work ahead, and our focus is on supporting points of vaccination , as that's key to increasing the volume of people getting vaccinated every day.#JPM2021 [Internet]. Twitter. 2021 [cited 2023 Jan 22]. Available from: https://twitter.com/pfizer/status/1349421959222853633
- 23. ModernaTx Inc. Your vaccination questions answered [Internet]. 2022 [cited 2023 Jan 22]. Available from: <u>http://web.archive.org/web/20230107040800/https://www.makeityourvaccine.com/faqs</u>
- 24. Pfizer. Sponsor briefing document [Internet]. 2021 Oct [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/153409/download
- 25. Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19 Vaccine for individuals 6 months through 4 years of age [Internet]. 2022 Jun [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/159393/download</u>
- 26. Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19

Vaccine for individuals 5 through 11 years of age [Internet]. 2021 Oct [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/153947/download

- 27. Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19 Vaccine for individuals 12 years of age and older [Internet]. 2021 May [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/148542/download</u>
- Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19 Vaccine booster dose [Internet]. 2021 Sep [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/152432/download</u>
- 29. Marks P. CBER assessment of Pfizer-BioNTech COVID-19 Vaccine second booster dose [Internet]. 2022 Mar [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/157364/download
- 30. Food and Drug Administration. FDA EUA review memorandum for Moderna COVID-19 Vaccine in individuals 6 months through 17 years of age [Internet]. 2022 Jun [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/159611/download
- 31. Food and Drug Administration. FDA EUA review memorandum for Moderna COVID-19 Vaccine booster dose [Internet]. 2021 Oct [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/153912/download
- 32. Marks P. CBER assessment of Moderna COVID-19 Vaccine second booster dose [Internet]. 2022 Mar [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/157368/download</u>
- 33. Food and Drug Administration. Transcript of 167th Vaccines and Related Biological Products Advisory Committee (VRBPAC) Meeting [Internet]. 2021 [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/154871/download</u>
- 34. Food and Drug Administration. Transcript of 172nd Vaccines and Related Biological Products Advisory Committee (VRBPAC) Meeting [Internet]. 2022 [cited 2023 Jan 22]. Available from: <u>https://www.fda.gov/media/158030/download</u>
- 35. Marks PW, Gruppuso PA, Adashi EY. Urgent Need for Next-Generation COVID-19 Vaccines. JAMA [Internet]. 2023 Jan 3;329(1):19–20. Available from: <u>http://dx.doi.org/10.1001/jama.2022.22759</u>
- 36. Food and Drug Administration. FDA EUA review memorandum for Pfizer-BioNTech COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5) for individuals 18 years of age and older [Internet]. 2022 Aug [cited 2023 Jan 24]. Available from: https://www.fda.gov/media/161595/download
- Food and Drug Administration. FDA EUA review memorandum for Moderna COVID-19 Vaccine, Bivalent (mRNA-1273.222) for individuals 18 years of age and older [Internet]. 2022 Aug [cited 2023 Jan 22]. Available from: https://www.fda.gov/media/161554/download
- 38. Pfizer Inc. Pfizer and BioNTech Announce Updated Clinical Data for Omicron BA.4/BA.5-Adapted Bivalent Booster Demonstrating Substantially Higher Immune Response in Adults Compared to the Original COVID-19 Vaccine [Internet]. 2022 [cited 2023 Jan 24]. Available from: <u>https://www.pfizer.com/news/press-release/press-release-detail/pfizer-and-biontech-announce-updated-clinical-data-omicron</u>

- 39. ModernaTx, Inc. Moderna's BA.4/BA.5 targeting bivalent booster, mRNA-1273.222, meets primary endpoint of superiority against Omicron variants compared to booster dose of mRNA-1273 in Phase 2/3 clinical trial [Internet]. 2022 [cited 2023 Jan 24]. Available from: https://investors.modernatx.com/news/news-details/2022/Modernas-BA.4BA.5-Targeting-Bivalent-Booster-mRNA-1273.222-Meets-Primary-Endpoint-of-Superiority-Against-Omicron-Variants-Compared-to-Booster-Dose-of-mRNA-1273-in-Phase-23-Clinical-Trial/default.aspx
- 40. ModernaTx Inc. EUA Fact Sheet for Recipients and Caregivers, Moderna COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5) for individuals 6 years of age and older [Internet]. 2022 [cited 2023 Jan 24]. Available from: <u>https://www.fda.gov/media/161318/download</u>
- Pfizer Inc. EUA Fact Sheet for Healthcare Providers, Pfizer-BioNTech COVID-19 Vaccine, Bivalent (Original and Omicron BA.4/BA.5), Gray Cap for individuals 12 years of age and older [Internet].
 2022. Available from: <u>https://www.fda.gov/media/161327/download</u>
- 42. Polack FP, Thomas SJ, Kitchin N, Absalon J, Gurtman A, Lockhart S, et al. Safety and Efficacy of the BNT162b2 mRNA Covid-19 Vaccine. N Engl J Med [Internet]. 2020 Dec 31;383(27):2603–15. Available from: <u>http://dx.doi.org/10.1056/NEJMoa2034577</u>
- Baden LR, El Sahly HM, Essink B, Kotloff K, Frey S, Novak R, et al. Efficacy and Safety of the mRNA-1273 SARS-CoV-2 Vaccine. N Engl J Med [Internet]. 2021 Feb 4;384(5):403–16. Available from: <u>http://dx.doi.org/10.1056/NEJMoa2035389</u>
- 44. Riley LE. mRNA Covid-19 Vaccines in Pregnant Women. N Engl J Med [Internet]. 2021 Jun 17;384(24):2342–3. Available from: <u>http://dx.doi.org/10.1056/NEJMe2107070</u>
- 45. Press briefing by White House COVID-19 response team and public health officials [Internet]. 2021 [cited 2023 Jan 22]. Available from: <u>https://youtu.be/vh-NbiGqtXw</u>
- 46. ACOG and SMFM recommend COVID-19 vaccination for pregnant individuals [Internet]. [cited 2023 Jan 5]. Available from: <u>https://www.acog.org/news/news-releases/2021/07/acog-smfm-recommend-covid-19-vaccination-for-pregnant-individuals</u>
- Pfizer Inc. Interim Report 6 Month Update: A Phase 1/2/3, Placebo-Controlled, Randomized, Observer-Blind, DoseFinding Study to Evaluate the Safety, Tolerability, Immunogenicity, and Efficacy of SARS-COV-2 RNA Vaccine Candidates Against COVID-19 in Healthy Individuals [Internet]. 2021 Apr. Available from: <u>https://clinical-information.canada.ca/ci-rc-vu.pdf?file=m5/53-clin-stud-rep/535-rep-effic-safety-stud/covid19/5351-stud-rep-contr/red-c4591001/red-c4591001-reportbody-2.pdf&id=252736
 </u>
- 48. Thomas SJ, Moreira ED Jr, Kitchin N, Absalon J, Gurtman A, Lockhart S, et al. Six month safety and efficacy of the BNT162b2 mRNA COVID-19 vaccine [Internet]. bioRxiv. medRxiv; 2021. Available from: <u>http://medrxiv.org/lookup/doi/10.1101/2021.07.28.21261159</u>
- 49. CDC Health Alert Network. Cases of Cerebral Venous Sinus Thrombosis with Thrombocytopenia after Receipt of the Johnson & Johnson COVID-19 Vaccine (CDCHAN-00442) [Internet]. 2021 [cited 2023 Jan 22]. Available from: https://emergency.cdc.gov/han/2021/han00442.asp
- 50. Lee G. COVID-19 Vaccine Safety Technical (VaST) Work Group [Internet]. Advisory Committee on Immunization Practices; 2021 Apr 23. Available from:

https://www.cdc.gov/vaccines/acip/meetings/downloads/slides-2021-04-23/05-COVID-Lee-508.pdf

- 51. Shimabukuro T. Thrombosis with thrombocytopenia syndrome (TTS) following Janssen COVID-19 vaccine [Internet]. Advisory Committee on Immunization Practices; 2021 Apr 23. Available from: https://www.cdc.gov/vaccines/acip/meetings/downloads/slides-2021-04-23/03-COVID-Shimabukuro-508.pdf
- Yousaf AR, Cortese MM, Taylor AW, Broder KR, Oster ME, Wong JM, et al. Reported cases of multisystem inflammatory syndrome in children aged 12-20 years in the USA who received a COVID-19 vaccine, December, 2020, through August, 2021: a surveillance investigation. Lancet Child Adolesc Health [Internet]. 2022 May;6(5):303–12. Available from: <u>http://dx.doi.org/10.1016/S2352-4642(22)00028-1</u>
- 53. Wong HL, Tworkoski E, Ke Zhou C, Hu M, Thompson D, Lufkin B, et al. Surveillance of COVID-19 vaccine safety among elderly persons aged 65 years and older. Vaccine [Internet]. 2022 Dec 1; Available from: http://dx.doi.org/10.1016/j.vaccine.2022.11.069
- Schneider J, Sottmann L, Greinacher A, Hagen M, Kasper HU, Kuhnen C, et al. Postmortem investigation of fatalities following vaccination with COVID-19 vaccines. Int J Legal Med [Internet]. 2021 Nov;135(6):2335–45. Available from: <u>http://dx.doi.org/10.1007/s00414-021-02706-9</u>
- 55. Sessa F, Salerno M, Esposito M, Di Nunno N, Zamboni P, Pomara C. Autopsy Findings and Causality Relationship between Death and COVID-19 Vaccination: A Systematic Review. J Clin Med Res [Internet]. 2021 Dec 15;10(24). Available from: <u>http://dx.doi.org/10.3390/jcm10245876</u>
- 56. Gill JR, Tashjian R, Duncanson E. Autopsy Histopathologic Cardiac Findings in 2 Adolescents Following the Second COVID-19 Vaccine Dose. Arch Pathol Lab Med [Internet]. 2022 Aug 1;146(8):925–9. Available from: <u>http://dx.doi.org/10.5858/arpa.2021-0435-SA</u>
- Schwab C, Domke LM, Hartmann L, Stenzinger A, Longerich T, Schirmacher P. Autopsy-based histopathological characterization of myocarditis after anti-SARS-CoV-2-vaccination. Clin Res Cardiol [Internet]. 2022 Nov 27;1–10. Available from: <u>http://dx.doi.org/10.1007/s00392-022-02129-5</u>
- 58. Safavi F, Gustafson L, Walitt B, Lehky T, Dehbashi S, Wiebold A, et al. Neuropathic symptoms with SARS-CoV-2 vaccination. medRxiv [Internet]. 2022 May 17; Available from: <u>http://dx.doi.org/10.1101/2022.05.16.22274439</u>
- 59. Kwan AC, Ebinger JE, Wei J, Le CN, Oft JR, Zabner R, et al. Apparent risks of postural orthostatic tachycardia syndrome diagnoses after COVID-19 vaccination and SARS-Cov-2 Infection. Nature Cardiovascular Research [Internet]. 2022 Dec 12 [cited 2023 Jan 5];1(12):1187–94. Available from: https://www.nature.com/articles/s44161-022-00177-8
- Blitshteyn S, Fedorowski A. The risks of POTS after COVID-19 vaccination and SARS-CoV-2 infection: it's worth a shot. Nature Cardiovascular Research [Internet]. 2022 Dec 12 [cited 2023 Jan 5];1(12):1119–20. Available from: <u>https://www.nature.com/articles/s44161-022-00180-z</u>
- 61. Gat I, Kedem A, Dviri M, Umanski A, Levi M, Hourvitz A, et al. Covid-19 vaccination BNT162b2 temporarily impairs semen concentration and total motile count among semen donors. Andrology

[Internet]. 2022 Sep;10(6):1016–22. Available from: http://dx.doi.org/10.1111/andr.13209

- 62. EMA. Meeting highlights from the Pharmacovigilance Risk Assessment Committee (PRAC) 24 27 October 2022 [Internet]. European Medicines Agency. 2022 [cited 2023 Jan 22]. Available from: <u>https://www.ema.europa.eu/en/news/meeting-highlights-pharmacovigilance-risk-assessment-</u> <u>committee-prac-24-27-october-2022</u>
- Edelman A, Boniface ER, Male V, Cameron ST, Benhar E, Han L, et al. Association between menstrual cycle length and covid-19 vaccination: global, retrospective cohort study of prospectively collected data. BMJ Med [Internet]. 2022 Sep 27;1(1). Available from: <u>http://dx.doi.org/10.1136/bmjmed-2022-000297</u>
- 64. Sheikh K. New Study Shows Covid Vaccines Can Temporarily Alter Menstrual Cycle. The New York Times [Internet]. 2022 Sep 27 [cited 2023 Jan 22]; Available from: <u>https://www.nytimes.com/2022/09/27/well/live/covid-vaccine-periods-delay.html</u>
- 65. Morris A. Women said coronavirus shots affect periods. New study shows they're right. The Washington Post [Internet]. 2022 Sep 27 [cited 2023 Jan 22]; Available from: https://www.washingtonpost.com/wellness/2022/09/27/covid-vaccine-period-late/
- Hanna N, Heffes-Doon A, Lin X, Manzano De Mejia C, Botros B, Gurzenda E, et al. Detection of Messenger RNA COVID-19 Vaccines in Human Breast Milk. JAMA Pediatr [Internet]. 2022 Dec 1;176(12):1268–70. Available from: <u>http://dx.doi.org/10.1001/jamapediatrics.2022.3581</u>
- 67. Knudsen B, Prasad V. COVID-19 vaccine induced myocarditis in young males: A systematic review. Eur J Clin Invest [Internet]. 2022 Dec 28;e13947. Available from: <u>http://dx.doi.org/10.1111/eci.13947</u>
- Mansanguan S, Charunwatthana P, Piyaphanee W, Dechkhajorn W, Poolcharoen A, Mansanguan C. Cardiovascular Manifestation of the BNT162b2 mRNA COVID-19 Vaccine in Adolescents. Trop Med Infect Dis [Internet]. 2022 Aug 19;7(8). Available from: <u>http://dx.doi.org/10.3390/tropicalmed7080196</u>
- Chua GT, Kwan MYW, Chui CSL, Smith RD, Cheung ECL, Ma T, et al. Epidemiology of Acute Myocarditis/Pericarditis in Hong Kong Adolescents Following Comirnaty Vaccination. Clin Infect Dis [Internet]. 2022 Sep 10;75(4):673–81. Available from: <u>http://dx.doi.org/10.1093/cid/ciab989</u>
- Kracalik I, Oster ME, Broder KR, Cortese MM, Glover M, Shields K, et al. Outcomes at least 90 days since onset of myocarditis after mRNA COVID-19 vaccination in adolescents and young adults in the USA: a follow-up surveillance study. Lancet Child Adolesc Health [Internet]. 2022 Nov;6(11):788–98. Available from: http://dx.doi.org/10.1016/S2352-4642(22)00244-9
- Fraiman J, Erviti J, Jones M, Greenland S, Whelan P, Kaplan RM, et al. Serious adverse events of special interest following mRNA COVID-19 vaccination in randomized trials in adults. Vaccine [Internet]. 2022 Sep 22;40(40):5798–805. Available from: <u>http://dx.doi.org/10.1016/j.vaccine.2022.08.036</u>